Author's response to reviews

Title: DCE-MRI assessment of the effect of Epstein-Barr Virus-encoded latent membrane protein-1 targeted DNAzyme on tumor vasculature in patients with nasopharyngeal carcinomas

Authors:

Wei-Hua Liao (ouwenliao@163.com)
Li-Fang Yang (yanglifang99@hotmail.com)
Xiao-Yu Liu (lxy9827@qq.com)
Gao-Feng Zhou (peak.zh@163.com)
Wu-Zhong Jiang (13973126486@189.cn)
Bob-Lei Hou (bhou@hsc.wvu.edu)
Lun-Quan Sun (lunquansun@csu.edu.cn)
Ya Cao (ycao98@vip.sina.com)
Xiao-Yi Wang (185223273@qq.com)

Version: 2
Date: 3 July 2014

Author's response to reviews:

1. Approval for this study was obtained from the Human Ethics Committee of the Xiangya Hospital, Central South University, and written consent was obtained from each participant. I have added this content in manuscript (line 196-198)
2. I have added Line Number in my paper.